\input{ThisWeek.tex}
\newcommand{\classday}{\Tuesday}
\input{../TemplateWeek/InClassWork.tex}
\pgfplotsset{compat=newest}
\input{Exercises.tex}

%\booltrue{teacher}

%Duplicated the question about tan, and it's too hard for the warm-up since they haven't necessarily seen it before


\section*{Warm Up}
\begin{enumerate}[label=(\roman*)]
\item \Exercise{E5-5_WarmUp1}
%\item \Exercise{E5-5_WarmUp2}
\item \Exercise{E5-6_WarmUp1}
\end{enumerate}

\section*{Exercise 1}
\Exercise{E5-5_Main1}

\section*{Exercise 2}
 \Exercise{E5-6_Main7}

\pagebreak

\section*{Exercise 3}
\Exercise{E5-5_Main2}

\section*{Exercise 4}
\Exercise{E5-5_Main3}

\section*{Exercise 5}
\Exercise{E5-5_Main4}

\section*{Exercise 6}
\Exercise{E5-5_Main5}

\pagebreak

\begin{itemize}
\item For Exercise 1, you should get $F(x)+C$ where $F(1)=\frac{-2}{3}$.
\item For 2a, about $2.2143$
\item For 2b, about $1.6094$
\item For 4, it's $-\ln|\cos x|$
\item For each of the parts of 6, you get functions $F_a(x)+C$, $F_b(x)+C$, and $F_c(x)+C$.  You should get $F_a(1)=e^e$, $F_b(1)=\tan^{-1}(e)$, and $F_c(1)=\frac{-2}{15}3^{3/2}$.
\end{itemize}


\end{document}